Author’s response to reviews

Title: Analysis of low birth weight and its co-variants in Bangladesh based on a sub-sample from nationally representative survey

Authors:

Jahidur Khan (jkhan@isrt.ac.bd)

Md. Islam (mislam6@isrt.ac.bd)

Nabil Awan (nawan@isrt.ac.bd)

Olav Muurlink (o.muurlink@gmail.com)

Version: 3 Date: 16 Feb 2018

Author’s response to reviews:

1) Ethical Approval

Please confirm whether your study was submitted to and approved by your institutional ethics committee and include a statement to this effect in your ethical approval and consent to participate section. Please also ensure that the full name of your ethics committee is included in this statement. If the need for ethics approval was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: Thanks for your comments. We have given detail explanation in “Ethics approval and consent to participate” subsection (page number 15).

2) Response Letter

Please remove the author response letter from the manuscript.

Response: We have addressed this issue.
3) Author Email

Please note that the email for author Olav Muurlink, differs between the submission system and the title page. Please edit this so the emails match.

Response: Thank you. We have provided the email of the author Olav Muurlink. Hope it will match now. Please check this and let us know if you find any problem.

4) Introduction Heading

Please edit the main text heading 'Introduction' to 'Background'.

Response: We have addressed this in Page 3.

5) List of Abbreviations

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Response: We have addressed this in Page 14.

6) Ethical Approval and consent to Participate Heading

Please edit the 'ethics' heading to read 'Ethical Approval and consent to Participate Heading'.

Response: We have addressed this in Page 15.
7) Author Contributions

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: We have addressed this in Page 15.

8) Funding Section

We note that you have not included a ‘Funding’ section in the Declarations. All sources of funding for the research reported should be declared. If no funding was obtained for your study we still require this section to be included with the statement “No funding was obtained for this study”.

Response: We have addressed this in Page 15.

9) Figure Legends

Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Response: We have addressed this in Page 21.
10) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have provided clean manuscript.